
•

• Stacking last week

•

• Red pbars collected for week.  
• Last week was 4th or 5th best week
• Stacking rate is down a little.
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• Stacking rate is down a little.

•

• Almost back to last year.
• Booster will try to get us back to 8e12 on target.

•

• Best hour of stacking and best 30 minutes.
• Slowly coming up, last three days in particular well.
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• Slowly coming up, last three days in particular well.

•

• Cumulative plot.
• Last year didn't quite match
• First two points for this year look good.

•

○ We have almost matched 2001 already!
○ Planned Pbar source in Germany is only 9500 on this plot.  Upgrade path is to double 

this.   2018 beam?
• Topics

• Operations
• Nearly back (await protons)
• Stacking 
• Transfers

• Issues(?)
• Emittances

□ Horizontal emittance was less than the vertical.   Vertical is still bigger, 
after fix.

□ The no beam response of emittance monitors is a few dB different than 
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after fix.
□ The no beam response of emittance monitors is a few dB different than 

in 2007
□ Also, DC offset on DCCT is much less. 
□ Both of those quantities go into transform.
□ Horizontal went up a lot.
□ Using A:BEAM
□ href=http://www-bd.fnal.gov/cgi-mach/machlog.pl?nb=pbar09

&action=view&page=247&anchor=100827&hilite=10:08:27-
• ARF4

□ Taking about a 1/3 more transfers.
□ What we see on the plot is a voltage jump.
□ The voltage request is low at the start.
□ The feedback is working, when the voltage is lower than we want, it 
□ Longitudinal emittance low on the transfers taking out.  We saw this 

before the AP0 wall current monitor scope died.
□ Access today to look at ARF4

□
• A:QT ground current (seepage onto ceramic?)
• D2A SEMs

□ Cant control gains remotely
□ Wants to try comparing old vs new on SEM806.

♦ Take some data.
♦ Histogram D44
♦ Do old vs new.
♦ We will do this next week.

• AP0 Wall Current Monitor
□ Scope died.
□ Prep receieved the scope.
□ The second scope is a different scope.
□ Optimized for old scope.

♦ Old TS680 there is a min vertical resolution.
♦ That is in the code.  That limit may not apply to the 3000 

series.
□ Set it for AC coupling.  Dave Peterson says this shouldn’t matter.
□ Somewhere there is a DC offset.
□ Bandwidth of scope is 1/2 of old scope.   This will mean proton torpedo.

• Acc BPMs 
□ Have to reboot them.
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□ Have to reboot them.
□ Can check with P207 and reboot any houses have problem

• Oscillation Overthruster
□ AP2 portion is now working again!
□ No longer have to target tune
□ Keith set orbit to best running conditions.
□ Had to change positions slightly at Septum magnet.

• A3B7
• Excursion frequency and amplitudes

□ same
• Tech Div.

□ Magnet is no test stand.
□ Main guy in charge in testing, Mark Thompson, is back from vacation 

and can start looking at this.
□ There are three sets of coils, there is confusion on the water 

connections.
• Worklist
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